Author's response to reviews

Title: Cellulite and extracorporeal Shockwave therapy (CelluShock-2009) - a Randomized Trial

Authors:

Karsten Knobloch (knobloch.karsten@mh-hannover.de)
Beatrice Joest (beatrice_Joest@web.de)
Peter M Vogt (vogt.peter@mh-hannover.de)

Version: 5 Date: 12 October 2010

Author's response to reviews:

Response to reviewers

Reviewer's report
Title: Cellulite and extracorporeal Shockwave therapy (CelluShock-2009) - a Randomized Trial
Version: 4 Date: 9 September 2010
Reviewer: Santiago Palacios
Reviewer's report:
The protocol is very well done and I have no objection to its publication.
The only point I find it interesting to be added, would be the differentiation between menopausal and postmenopausal patients in terms of results, and within the premenopausal group, the identification of those who take oral contraceptives.
Level of interest: An article whose findings are important to those with closely related research interests
Quality of written English: Acceptable
Statistical review: No, the manuscript does not need to be seen by a statistician.
Declaration of competing interests:
I declare that I have no competing interests.

Dear reviewer,

thank you very much for your kind consideration of our work.

We added your suggestion regarding pre- and postmenopausal women and the associated medication, which is documented, however, we did not comment on this in detail in the study plan. As such, we added: “Both, pre- and postmenopausal women are included with documentation of their status and
concomitant medication." in the participants section.
Furthermore, we deleted some spelling errors throughout the manuscript.
Sincerely yours
Karsten Knobloch, FACS